Author’s response to reviews

Title: Association of Photopic and Mesopic Contrast Sensitivity in Older Drivers with Risk of Motor Vehicle Collision Using Naturalistic Driving Data

Authors:

Cynthia Owsley (owsley@uab.edu)
Thomas Swain (thomasswain@uabmc.edu)
Rong Liu (liur@uab.edu)
Gerald McGwin (gmcgwin@uabmc.edu)
MiYoung Kwon (miyoungkwon@uabmc.edu)

Version: 2 Date: 02 Jan 2020

Author’s response to reviews:

Dear Dr. Papageorgiou

Thank you for considering this manuscript. We look forward to hearing from you regarding the revised submission. We have revised the manuscript and have also attached a response to reviewer comments for the 2nd resubmission. Both have been uploaded.

Thank you very much.

Sincerely,

Cynthia Owsley (corresponding author)
on behalf of all authors